Author's response to reviews

Title: Using reflexivity to enhance in-depth interviewing skills for the clinician researcher.

Authors:

Ruth P McNair (r.mcnaire@unimelb.edu.au)
Angela Taft (a.taft@latrobe.edu.au)
Kelsey Hegarty (k.hegarty@unimelb.edu.au)

Version: 3 Date: 24 September 2008

Author's response to reviews: see over
24 September 2008

The Editor
BMC Medical Research Methodology

Regarding paper for consideration for publication:
Using reflexivity to enhance in-depth interviewing skills for the clinician researcher.

Dear Editor,

Thank you for accepting our paper for publication in BMC Medical Research Methodology.
We have responded to the minor suggestions of the reviewer Karin Reid. These include deleting the ethics reference number, clarifying who ‘this GP researcher’ was referring to in the opening of the discussion, adding a subtitle to the section on reflexivity, and amending the sentence introducing our recommendations.

We have also ensured that the paper follows the required format.

Yours sincerely

Ruth McNair
Angela Taft
Kelsey Hegarty